Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “A Case Report of the First Platelet Transfusion Refractoriness in a Patient with Acute Myelocytic Leukemia” (ID: 86455). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as flowing: Responds to the reviewer’s comments:

Reviewer1
1. Could you supplement the platelet antibody detection of this patient at the first visit?
   **Response:** The five items of anti-platelet antibody testing were improved positive anti-platelet membrane glycoprotein IIb and IIIa specific antibodies.
2. Does the patient have a family history of immune-associated thrombocytopenia?
   **Response:** There was no family history of immune-associated thrombocytopenia.

Reviewer2:
1. Could you supplement the platelet antibody detection of this patient at the first visit?
   **Response:** The five items of anti-platelet antibody testing were improved positive anti-platelet membrane glycoprotein IIb and IIIa specific antibodies.
2. Does the patient have a family history of immune-associated thrombocytopenia?
   **Response:** There was no family history of immune-associated thrombocytopenia.